Author’s response to reviews

Title: Characterization of Non-Tuberculous Mycobacterium from Humans and Water in an Agro pastoral area in Zambia

Authors:

Sydney Malama (sydneymalama1971@gmail.com)
Ngula Monde (nkabelenga@yahoo.co.uk)
Musso Munyeme (mussomunyeme@gmail.com)
Adrian Muwonge (adrian.muwonge@roslin.ed.ac.uk)
John Muma (jbwalya@lycos.com)

Version: 1 Date: 09 Oct 2017

Author’s response to reviews:

Dear Editor

Thanks for your quick feedback. We wish to inform you that we have addressed the following based on your feedback:

1. Declarations: We have added the section on all the required declarations
2. We have added the names and details of the authors in the subsystem
3. We have included all mandatory sections and where necessary we have indicated "not applicable"
4. We have further cleaned the manuscript for grammatical errors
5. Author information is also added.

Regards

Dr Sydney Malama, corresponding author